\documentclass[CRMATH,Unicode,biblatex,XML]{cedram}

\TopicFR{Théorie des nombres}
\TopicEN{Number theory}

\addbibresource{CRMATH_Aymone_20230928.bib}

\usepackage{dsfont}
\usepackage{amssymb}
\usepackage{mathtools}
\usepackage{enumerate}

\newcommand{\ind}{\mathds{1}}
\renewcommand{\Re}{\mathrm{Re}}




%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand{\dd}{\mathrm{d}}

%%%%%%%%% a placer avant \begin{document}
\begin{thebibliography}{10}

\bibitem{xu_positivity}
{\sc R.~Angelo and M.~Xu}, {\em On a {T}ur\'an conjecture and random
 multiplicative functions}, Quart. J. Math., online first (2022).

\bibitem{aymone_JLMS}
{\sc M.~Aymone, W.~Heap, and J.~Zhao}, {\em Partial sums of random
 multiplicative functions and extreme values of a model for the {R}iemann zeta
 function}, J. Lond. Math. Soc. (2), 103 (2021), pp.~1618--1642.

\bibitem{aymone_sign_changes}
\leavevmode\vrule height 2pt depth -1.6pt width 23pt, {\em Sign changes of the
 partial sums of a random multiplicative function}, Bull. Lond. Math. Soc., 55
 (2023), pp.~78--89.

\bibitem{aymonebiased}
{\sc M.~Aymone and V.~Sidoravicius}, {\em Partial sums of biased random
 multiplicative functions}, J. Number Theory, 172 (2017), pp.~343--382.

\bibitem{basquinn}
{\sc J.~Basquin}, {\em Sommes friables de fonctions multiplicatives
 al\'eatoires}, Acta Arith., 152 (2012), pp.~243--266.

\bibitem{benatar_rodgers}
{\sc J.~Benatar, A.~Nishry, and B.~Rodgers}, {\em Moments of polynomials with
 random multiplicative coefficients}, Mathematika, 68 (2022), pp.~191--216.

\bibitem{seip_helsons}
{\sc A.~Bondarenko and K.~Seip}, {\em Helson's problem for sums of a random
 multiplicative function}, Mathematika, 62 (2016), pp.~101--110.

\bibitem{rachidupperbound}
{\sc R.~Caich}, {\em Almost sure upper bound for random multiplicative functions}, preprint available at arXiv:2304.00943, (2023).


\bibitem{chatterjeemult}
{\sc S.~Chatterjee and K.~Soundararajan}, {\em Random multiplicative functions
 in short intervals}, Int. Math. Res. Not. IMRN, (2012), pp.~479--492.

\bibitem{erdosuns}
{\sc P.~Erd\H{o}s}, {\em Some unsolved problems}, Magyar Tud. Akad. Mat.
 Kutat\'o Int. K\"ozl., 6 (1961), pp.~221--254.

\bibitem{halasz}
{\sc G.~Hal{\'a}sz}, {\em On random multiplicative functions}, in Hubert
 {D}elange colloquium ({O}rsay, 1982), vol.~83 of Publ. Math. Orsay, Univ.
 Paris XI, Orsay, 1983, pp.~74--96.

\bibitem{hardy_weighted}
{\sc S.~ Hardy}, {\em Almost sure bounds for a weighted {S}teinhaus random multiplicative function}, preprint available at arXiv:2307.00499, (2023).

\bibitem{harpergaussian}
{\sc A.~J. Harper}, {\em Bounds on the suprema of {G}aussian processes, and
 omega results for the sum of a random multiplicative function}, Ann. Appl.
 Probab., 23 (2013), pp.~584--616.

\bibitem{harpermult}
\leavevmode\vrule height 2pt depth -1.6pt width 23pt, {\em On the limit
 distributions of some sums of a random multiplicative function}, J. Reine
 Angew. Math., 678 (2013), pp.~95--124.

\bibitem{harper_high_moments}
\leavevmode\vrule height 2pt depth -1.6pt width 23pt, {\em Moments of random
 multiplicative functions, {II}: {H}igh moments}, Algebra Number Theory, 13
 (2019), pp.~2277--2321.

\bibitem{harperfirstmoment}
\leavevmode\vrule height 2pt depth -1.6pt width 23pt, {\em Moments of random
 multiplicative functions, {I}: {L}ow moments, better than squareroot
 cancellation, and critical multiplicative chaos}, Forum Math. Pi, 8 (2020),
 pp.~e1, 95.

\bibitem{harper_largefluctations}
\leavevmode\vrule height 2pt depth -1.6pt width 23pt, {\em Almost sure large
 fluctuations of random multiplicative functions}, Int. Math. Res. Not.,
 (2022).

\bibitem{harper_ashkhan}
{\sc A.~J. Harper, A.~Nikeghbali, and M.~Radziwi\l\l}, {\em A note on
 {H}elson's conjecture on moments of random multiplicative functions}, in
 Analytic number theory, Springer, Cham, 2015, pp.~145--169.

\bibitem{haselgrove}
{\sc C.~B. Haselgrove}, {\em A disproof of a conjecture of {P}\'{o}lya}, 
Mathematika, 5 (1958), pp.~141--145.

\bibitem{heap_lindqvist}
{\sc W.~P. Heap and S.~Lindqvist}, {\em Moments of random multiplicative
 functions and truncated characteristic polynomials}, Q. J. Math., 67 (2016),
 pp.~683--714.

\bibitem{houghmult}
{\sc B.~Hough}, {\em Summation of a random multiplicative function on numbers
 having few prime factors}, Math. Proc. Cambridge Philos. Soc., 150 (2011),
 pp.~193--214.

\bibitem{humphries_sign_changes}
{\sc P.~Humphries}, {\em The distribution of weighted sums of the {L}iouville
 function and {P}\'{o}lya's conjecture}, J. Number Theory, 133 (2013),
 pp.~545--582.

\bibitem{kalmynin_quadratic}
{\sc A.~Kalmynin}, {\em Quadratic characters with positive partial sums}, Mathematika, 69 (2023),
 pp.~90--99.


\bibitem{klurman_positivity}
{\sc B.~Kerr and O.~Klurman}, {\em How negative can $\sum_{n\leq
 x}\frac{f(n)}{n}$ be?}, arXiv:2211.05540, (2022).

\bibitem{klurman_rand_chowla}
{\sc O.~Klurman, I.~D. Shkredov, and M.~W. Xu}, {\em On the random {C}howla
 conjecture}, GAFA (to appear), (2023).

\bibitem{tenenbaum2013}
{\sc Y.-K. Lau, G.~Tenenbaum, and J.~Wu}, {\em On mean values of random
 multiplicative functions}, Proc. Amer. Math. Soc., 141 (2013), pp.~409--420.

\bibitem{levy}
{\sc P.~L\'evy}, {\em Sur les s\'eries dont les termes sont des variables
 \'eventuelles ind\'ependents}, Studia Mathematica, 03 (1931), pp.~119--155.

\bibitem{trudgian_sign_changes}
{\sc M.~J. Mossinghoff and T.~S. Trudgian}, {\em Between the problems of
 {P}\'{o}lya and {T}ur\'{a}n}, J. Aust. Math. Soc., 93 (2012), pp.~157--171.

\bibitem{najnudel}
{\sc J.~Najnudel}, {\em On consecutive values of random completely
 multiplicative functions}, Electron. J. Probab., 25 (2020), pp.~Paper No. 59,
 28.

\bibitem{xu_clt}
{\sc K.~Soundararajan and M.~Xu}, {\em Central limit theorems for random
 multiplicative functions}, arXiv:2212.06098, (2022).

\bibitem{wintner}
{\sc A.~Wintner}, {\em Random factorizations and {R}iemann's hypothesis}, Duke
 Math. J., 11 (1944), pp.~267--275.

\bibitem{xu_firstmoment}
{\sc M.~W. Xu}, {\em Better than square-root cancellation for random
 multiplicative functions}, arXiv:2303.06774, (2023).

\end{thebibliography}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand*{\mk}{\mkern -1mu}
\newcommand*{\Mk}{\mkern -2mu}
\newcommand*{\mK}{\mkern 1mu}
\newcommand*{\MK}{\mkern 2mu}

\hypersetup{urlcolor=purple, linkcolor=blue, citecolor=red}


\newcommand*{\romanenumi}{\renewcommand*{\theenumi}{\roman{enumi}}}
\newcommand*{\Romanenumi}{\renewcommand*{\theenumi}{\Roman{enumi}}}
\newcommand*{\alphenumi}{\renewcommand*{\theenumi}{\alph{enumi}}}
\newcommand*{\Alphenumi}{\renewcommand*{\theenumi}{\Alph{enumi}}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%% Auteur

\author{\firstname{Marco} \lastname{Aymone}}

\address{Departamento de Matem\'atica, Universidade Federal de Minas Gerais (UFMG), Av.~Antônio Carlos, 6627, CEP 31270-901, Belo Horizonte, MG, Brazil} 

\email{aymone.marco@gmail.com}

\thanks{I am thankful the CNPq for supporting this visit at Aix-Marseille Université -- grant PDE no.~400010/2022-4 (200121/2022-7). This project also was funded by CNPq grant Universal no.~403037/2021-2 and by FAPEMIG grant Universal no. APQ-00256-23.}

\CDRGrant[PDE]{400010/2022-4 (200121/2022-7)}
\CDRGrant[CNPq]{403037/2021-2}
\CDRGrant[FAPEMIG]{APQ-00256-23}

%%%%% Sujet


\altkeywords{\kwd{Fonctions multiplicatives aléatoires}\kwd{théorèmes d'oscillation}}

\keywords{\kwd{Random multiplicative functions}\kwd{Oscillation theorems}}

\subjclass{11K65, 11N99}

%%%%% Gestion


%%%%% Titre et résumé

\title{Sign changes of the partial sums of a random multiplicative function II}


\alttitle{Changements de signe des sommes partielles d'une fonction multiplicative aléatoire II}

\begin{abstract}
We study two models of random multiplicative functions: Rademacher random multiplicative functions supported on the squarefree integers $f$, and Rademacher random completely multiplicative functions $f^*$. We prove that the partial sums $\sum_{n\leq x}f^*(n)$ and $\sum_{n\leq x}\frac{f(n)}{\sqrt{n}}$ change sign infinitely often as $x\to\infty$, almost surely. The case $\sum_{n\leq x}\frac{f^*(n)}{\sqrt{n}}$ is left as an open question and we stress the possibility of only a finite number of sign changes, with positive probability. 
\end{abstract}

\begin{altabstract}
Nous étudions deux modèles de fonctions multiplicatives aléatoires : les fonctions multiplicatives aléatoires de Rademacher supportées sur les entiers sans carrés $f$, et les fonctions multiplicatives aléatoires complètement multiplicatives de Rademacher $f^*$. Nous prouvons que les sommes partielles $\sum_{n\leq x}f^*(n)$ et $\sum_{n\leq x}\frac{f(n)}{\sqrt{n}}$ changent de signe infiniment souvent comme $x\to\infty$, presque sûrement. Le cas $\sum_{n\leq x}\frac{f^*(n)}{\sqrt{n}}$ reste une question ouverte et nous soulignons la possibilité de seulement un nombre fini de changements de signe, avec probabilité positive.
\end{altabstract}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%








\begin{document}

\maketitle
\section{Introduction}
\subsection{Main results and motivation} A Rademacher random multiplicative function $f$ is defined as follows: Over the primes, the values $f(p)$ are \textit{i.i.d.} random variables taking $\pm 1$ values with probability $1/2$ in each instance (Rademacher random variables), and at the other values of $n$,
\[
f(n)\coloneqq \mu^2(n)\prod_{p|n}f(p),
\]
where $\mu$ is the M\"obius function.

By a Rademacher random completely multiplicative function $f^*$, we mean that
\[
f^*(n)=\prod_{p^a\| n}f(p)^a.
\]

In a previous paper, Aymone, Heap and Zhao~\cite{aymone_sign_changes} proved that the partial sums $\sum_{n\leq x}f(n)$ of a Rademacher random multiplicative function change sign infinitely often\footnote{Here we say that a function $M(x)$ changes sign infinitely often as $x\to\infty$ if at a certain point $x_0$, $M(x)$ is not always non-negative or non-positive for all $x\geq x_0$.} as $x\to\infty$, almost surely. The key idea was to show that, almost surely
\begin{align*}
\lim_{\sigma\to1/2^+}\int_1^\infty\left(\sum_{n\leq x}f(n)\right)\frac{\dd x}{x^{1+\sigma}}=0,\\
\lim_{\sigma\to1/2^+}\int_1^\infty\left|\sum_{n\leq x}f(n)\right|\frac{\dd x}{x^{1+\sigma}}=\infty,
\end{align*} 
and these visibly capture an infinite number of sign changes.

Our task here is to study sign changes for weighted sums of $f$ and $f^*$:
\[
\sum_{n\leq x}\frac{f(n)}{n^\alpha},\,\sum_{n\leq x}\frac{f^{*}(n)}{n^\alpha}, \;\alpha\geq 0.
\]
In both cases, we have that for $\alpha>1/2$, the partial sums converge to a non-vanishing Euler product, and hence, it can be shown that these partial sums become positive for all $x$ sufficiently large, almost surely. Therefore, we restrict ourselves to the cases $0\leq\alpha\leq 1/2$.

Our motivation comes from the study of the sign changes of the partial sums of the deterministic counterparts to $f$ and $f^*$: the M\"obius $\mu$ and the Liouville $\lambda$. For $0\leq \alpha <1/2$, the partial sums $\sum_{n\leq x}\mu(n)n^{-\alpha}$ change sign infinitely often as $x\to\infty$, due to the analytic properties of $1/\zeta(s)$ and to the Landau's oscillation theorem. Answering a question of P\'olya, Haselgrove~\cite{haselgrove} proved that $\sum_{n\leq x}\lambda(n)$ changes sign infinitely often as $x\to\infty$. In the range $0<\alpha<1/2$, the case of an infinite number of sign changes of $\sum_{n\leq x}\lambda(n)n^{-\alpha}$ recasts questions about the non-trivial Riemann zeros, and at the present moment, unconditional results are unknown, see~\cite{trudgian_sign_changes} by Mossinghoff and Trudgian. In a paper by Humphries~\cite{humphries_sign_changes}, it can be deduced from his results that under RH and other hypothesis on Riemann zeros, the partial sums $\sum_{n\leq x}\lambda(n)n^{-\alpha}$ change sign infinitely often as $x\to\infty$. 

In both cases $\sum_{n\leq x}\mu(n)n^{-1/2}$ and $\sum_{n\leq x}\lambda(n)n^{-1/2}$ the problem of an infinitude of sign changes seems to be intricate and even assuming standard hypothesis, conditional results are unknown. In~\cite{trudgian_sign_changes} is left as an open problem to prove or disprove that $\sum_{n\leq x}\lambda(n)n^{-1/2}$ is negative for all $x\geq 17$. A remarkable consequence of this fact is that this implies that all non-trivial zeros of the Riemann zeta function have multiplicity at most $2$, see~\cite[Theorem~2.4]{trudgian_sign_changes}. In~\cite{humphries_sign_changes} it is showed that under standard assumptions on the non-trivial zeros of the Riemann zeta function, the set of $x$ for which $\sum_{n\leq x}\frac{\lambda(n)}{\sqrt{n}}$ is negative has total logarithmic asymptotic density.

Taking all of these facts into account, for the random counterparts $f$ and $f^*$ we were able to proof the following results.

\begin{theorem}\label{teorema principal 1} Let $f$ be a Rademacher random multiplicative function. Then for each $0\leq\alpha\leq 1/2$, $\sum_{n\leq x}\frac{f(n)}{n^\alpha}$ changes sign infinitely often as $x\to\infty$, almost surely.
\end{theorem} 
\begin{theorem}\label{teorema principal 2}
Let $f^*$ be a Rademacher random completely multiplicative function. Then for each $0\leq\alpha< 1/2$ we have that $\sum_{n\leq x}\frac{f^*(n)}{n^\alpha}$ changes sign infinitely often as $x\to\infty$, almost surely.
\end{theorem}

The case $\sum_{n\leq x}\frac{f^*(n)}{\sqrt{n}}$ is left as an open question, and it is possible that these sums change sign only a finite number of times with positive probability, here we give an heuristic reason. In a recent paper, Angelo and Xu~\cite{xu_positivity} (and later improvements by Kerr and Klurman~\cite{klurman_positivity}) studied the weighted sums $\sum_{n\leq x}\frac{f^*(n)}{n}$ in the context of a ``\textit{random Turán conjecture}''. They proved that the probability of $\sum_{n\leq x}\frac{f^*(n)}{n}$ being positive for all $x\geq1$ is at least $1-10^{-45}$. This implies a certain positivity for harmonic sums of $f^*$ and it may be possible that such result may be extended for $\sum_{n\leq x}\frac{f^*(n)}{n^\sigma}$, where $\sigma>1/2$. Further, after a convolution argument, $f^*=f\ast \ind_{PS}$, where $\ind_{PS}(n)$ is the indicator that $n$ is a perfect square. Therefore, 
\[
\sum_{n\leq x}\frac{f^*(n)}{\sqrt{n}}\approx \sum_{n\leq x}\frac{f(n)}{\sqrt{n}}\log(x/n).
\]
The later is known as a Riesz mean, and at some instances these averages stabilizes oscillatory properties of an oscillating arithmetic function providing a finite number of sign changes for these averages. 

 \subsection{Proof method} Our proof follows closely the lines of~\cite{aymone_sign_changes}. The difference here is that, in contrast with unweighted sums of a Rademacher random multiplicative function, we have the almost sure limits
\begin{align*}
\lim_{\sigma\to1/2^+}\int_1^\infty\left(\sum_{n\leq x}\frac{f(n)}{\sqrt{n}}\right)\frac{\dd x}{x^{1/2+\sigma}}=\infty,\\
\lim_{\sigma\to1/2^+}\int_1^\infty\left|\sum_{n\leq x}\frac{f(n)}{\sqrt{n}}\right|\frac{\dd x}{x^{1/2+\sigma}}=\infty,
\end{align*} 
and this alone does not capture an infinite number of sign changes, unless we make the two divergences above quantitative, and that these divergences differ in size. This can, indeed, be achieved by using the following quantitative statement by Harper:

\begin{theorem}[{Harper~\cite[p.~25]{harpergaussian}}]\label{teorema Harper} Let $p$ run over the primes and $\big(f(p)\big)_p$ be a sequence of independent Rademacher random variables. For fixed $A>3$, almost surely, there exists a sequence $\sigma_k\to1/2^+$ such that
\begin{align*}\label{equacao harper}
\sup_{1\leq t\leq 2\left(\log\left(\frac{1}{\sigma_k-1/2}\right)\right)^2}\Mk \left(\sum_{p}\frac{f(p)\cos(t\log p)}{p^{\sigma_k}}\Mk-\Mk2\log\log\left(\frac{1}{\sigma_k\Mk-\Mk1/2}\right)\right) %\\ 
\geq\log\left(\frac{1}{\sigma_k\Mk-\Mk1/2}\right)-A\log\log\left(\frac{1}{\sigma_k\Mk-\Mk 1/2}\right).
\end{align*} 
\end{theorem}

\subsection{Background} Here we are intended to give a list of results in the literature (perhaps non-complete) on random multiplicative functions. The first result on partial sums of random multiplicative functions is due to Wintner~\cite{wintner}, where he considered a question by Levy~\cite{levy} and showed that $\sum_{n\leq x}f(n)$ is almost surely $\Omega(x^{1/2-\epsilon})$ and $\ll x^{1/2+\epsilon}$, for any $\epsilon>0$.

Later, these results have been improved by Erd\H{o}s~\cite{erdosuns} and subsequently by Hal\'asz~\cite{halasz}. Later Basquin~\cite{basquinn} and Lau, Tenenbaum and Wu~\cite{tenenbaum2013} made a significant improvement by proving that, almost surely, 
\[\postdisplaypenalty1000000
\sum_{n\leq x}f(n)\ll \sqrt{x}(\log\log x)^{2+\epsilon},
\]
for any $\epsilon>0$.

In this year of 2023, Caich~\cite{rachidupperbound} proved that we can replace $(\log \log x)^{2+\epsilon}$ above by $(\log \log x)^{1/4+\epsilon}$. This combined with Harper's Omega bound~\cite{harper_largefluctations} gives a sharp description of the size of the fluctuations of the partial sums of $f$: 
\[
\sum_{n\leq x}f(n)=\Omega(\sqrt{x}(\log\log x)^{1/4-\epsilon}),
\] 
for any $\epsilon>0$, almost surely. 

Central limit Theorems also have been studied in various settings:~\cite{chatterjeemult, harpermult, houghmult, klurman_rand_chowla, xu_clt}.

The problem of moments also have been extensively studied, with the remarkable result of Harper that shows that the first moment exhibits better than squareroot cancellation~\cite{harperfirstmoment}. Other related works (including related models):
~\cite{aymone_JLMS, benatar_rodgers, seip_helsons, hardy_weighted, harper_high_moments, harper_ashkhan, heap_lindqvist, xu_firstmoment}.
And another viewpoints of investigation~\cite{xu_positivity, aymone_sign_changes, aymonebiased, kalmynin_quadratic, klurman_positivity, najnudel}.



\section{Preliminaries}
\subsection{Notation} Here $p$ stands for a generic prime number. We use the standard Vinogradov notation $f(x)\ll g(x)$ or Landau's $f(x)=O(g(x))$ whenever there exists a constant $c>0$ such that $|f(x)|\leq c|g(x)|$, for all $x$ in a set of parameters. When not specified, this set of parameters is an infinite interval $(a,\infty)$ for sufficiently large $a>0$. The standard $f(x)=o(g(x))$ means that $f(x)/g(x)\to0$ when $x\to a$, where $a$ could be a complex number or $\pm \infty$. For $g(x)>0$ for all $x$, we say that $f(x)=\Omega(g(x))$ if $\limsup_{x\to\infty}\frac{|f(x)|}{g(x)}>0$. 
\subsection{Some Lemmas}
By the pioneering work of Wintner~\cite{wintner}, we have that $F(s)\coloneqq \sum_{n=1}^\infty\frac{f(n)}{n^s}$ converges for all $\Re(s)>1/2$, almost surely. His proof can be modified to show that the same is true for $F^*(s)=\sum_{n=1}^\infty\frac{f^*(n)}{n^s}$. Moreover we have:
\begin{lemma}\label{lemma Euler product} For all $\Re(s)>1/2$, almost surely we have
\begin{align*}
F(s)&=\prod_{p}\left(1+\frac{f(p)}{p^s}\right),\\
F^*(s)&=\prod_{p}\left(1-\frac{f(p)}{p^s}\right)^{-1}.
\end{align*}
\end{lemma}

This leads to the almost sure identities
\begin{align}\label{equacao exponencial F}
F(s)&=\exp\left(\sum_{p}\frac{f(p)}{p^s}-\frac{1}{2}\log\zeta(2s)+O(1)\right),
\\
\label{equacao exponencial F*}
F^*(s)&=\exp\left(\sum_{p}\frac{f(p)}{p^s}+\frac{1}{2}\log\zeta(2s)+O(1)\right),
\end{align}
where the $O(1)$ term above is actually an analytic function that is uniformly bounded in $\Re(s)\geq 1/3+\epsilon$, for small fixed $\epsilon>0$. For a proof of these results we refer reader to~\cite[Lemma~2.4]{aymone_sign_changes} and the references therein.
To conclude this section, we have
\begin{lemma}\label{lemma upper bounding RDS}
Let $(f(p))_p$ be independent Rademacher random variables. Then, for all $\epsilon>0$, we have almost surely as $\sigma\to1/2^+$
\[
\sum_p\frac{f(p)}{p^\sigma}\ll \left(\log\left(\frac{1}{\sigma-1/2}\right)\right)^{1/2+\epsilon}.
\]
\end{lemma}
For a proof we refer reader to~\cite[Lemma~2.1 and Remark~2.1]{aymone_sign_changes}. 
\section{Proof of the main results}
\subsection{The case \texorpdfstring{$\sum_{n\leq x}\frac{f(n)}{\sqrt{n}}$}{sum l <= x (F(n))/(sqrt n)}} By the upper bound for the partial sums of a Rademacher random multiplicative function~\cite{basquinn} and~\cite{tenenbaum2013},
\[
\sum_{n\leq x}f(n)\ll \sqrt{x}(\log\log x)^{2+\epsilon},
\]
we have that by partial summation
\[
\sum_{n\leq x}\frac{f(n)}{\sqrt{n}}\ll (\log x)^2, \;a.s.
\]
Further refinements of this last upper bound could, perhaps, be achieved, by following the lines of~\cite{aymone_JLMS} and more recently of~\cite{hardy_weighted}. 

Now, for $\Re(s)=\sigma>1/2$,
\[
F(s)\coloneqq \sum_{n=1}^\infty \frac{f(n)}{n^s}=\sum_{n=1}^\infty \frac{f(n)}{\sqrt{n}}\frac{1}{n^{s-1/2}}=(s-1/2)\int_{1}^\infty \left(\sum_{n\leq x}\frac{f(n)}{\sqrt{n}} \right)\frac{\dd x}{x^{s+1/2}}.
\]

By equation~\eqref{equacao exponencial F} and Lemma~\ref{lemma upper bounding RDS} combined with the estimate $\log\zeta(2\sigma)=\log(1/(2\sigma-1))+O(1)$ valid for $\sigma\to 1/2^+$, we obtain that as $\sigma\to 1/2^+$, almost surely we have $F(\sigma)= (2\sigma-1)^{1/2+o(1)}$.
Hence, we almost surely have as $\sigma\to1/2^+$
\begin{equation}\label{equacao integral f rad weighted}
\int_{1}^\infty \left(\sum_{n\leq x}\frac{f(n)}{\sqrt{n}} \right)\frac{\dd x}{x^{\sigma+1/2}}=\frac{1}{(2\sigma-1)^{1/2+o(1)}}.
\end{equation} 

On the other hand, for $t\geq 1$
\[
\left|\frac{F(\sigma+it)}{t}\right|\ll \int_{1}^\infty \left|\sum_{n\leq x}\frac{f(n)}{\sqrt{n}} \right|\frac{\dd x}{x^{\sigma+1/2}}.
\]

Now, by equation~\eqref{equacao exponencial F},
\[
|F(s)|=\exp\left(\sum_{p}\frac{f(p)\cos(t\log p)}{p^\sigma}-\frac{1}{2}\Re(\log\zeta(2s))+O(1)\right).
\]

We recall that $|\log \zeta(\sigma+it)|\leq \log\log|t|+O(1)$, for $\sigma\geq 1$ and $t\geq 2$. By Theorem~\ref{teorema Harper}, almost surely we have an infinite sequence of points $\sigma_k\to1/2^+$ and $t_k\geq 1$ such that
\[
\left|\frac{F(\sigma_k+it_k)}{t_k}\right|\geq\exp\left((1+o(1))\log\left(\frac{1}{\sigma_k-1/2}\right)\right)\gg \frac{1}{(\sigma_k-1/2)^{0.9}}.
\]
Therefore, almost surely, there exists a sequence $\sigma_k\to1/2^+$ such that
\begin{equation}\label{equacao divergencia modulo rad weighted}
\int_{1}^\infty \left|\sum_{n\leq x}\frac{f(n)}{\sqrt{n}} \right|\frac{\dd x}{x^{\sigma_k+1/2}}\gg \frac{1}{(\sigma_k-1/2)^{0.9}}.
\end{equation}

To complete the argument, we see that the quantities at the left of~\eqref{equacao integral f rad weighted} and~\eqref{equacao divergencia modulo rad weighted} have different size, and so the partial sums $\sum_{n\leq x}\frac{f(n)}{\sqrt{n}}$ cannot be always non-negative or non-positive, there always must be infinite sign changes, almost surely.

\subsection{The case \texorpdfstring{$\sum_{n\leq x}\frac{f^*(n)}{n^\alpha}$}{sum n <= x (f*n)/(n alpha)}} Let then $0\leq\alpha<1/2$. Let $F^*(s)$ be the Dirichlet series of $f^*$. Similarly as in the previous case, by equation~\eqref{equacao exponencial F*} and Lemma~\ref{lemma upper bounding RDS}, we have that
\[
F^*(\sigma)=\frac{1}{(2\sigma-1)^{1/2+o(1)}},
\]
almost surely as $\sigma\to1/2^+$.

By partial summation and integration,
\[
F^*(s)=(s-\alpha)\int_1^\infty \left(\sum_{n\leq x}\frac{f^*(n)}{n^\alpha}\right)\frac{\dd x}{x^{s+1-\alpha}}.
\]
Therefore, almost surely as $\sigma\to 1/2^+$
\begin{equation}\label{equacao rad compl diverg sem mod}
\int_1^\infty \left(\sum_{n\leq x}\frac{f^*(n)}{n^\alpha}\right)\frac{\dd x}{x^{\sigma+1-\alpha}}=\frac{1+o(1)}{(1/2-\alpha)}\frac{1}{(2\sigma-1)^{1/2+o(1)}}.
\end{equation}
And similarly to~\eqref{equacao divergencia modulo rad weighted}, almost surely there exists a sequence $\sigma_k\to1/2^+$ such that
\begin{equation}\label{equacao rad compl diverg mod}
\int_1^\infty \left|\sum_{n\leq x}\frac{f^*(n)}{n^\alpha}\right|\frac{\dd x}{x^{\sigma+1-\alpha}}\gg\frac{1}{(2\sigma_k-1)^{0.99}}.
\end{equation}
Just as before,~\eqref{equacao rad compl diverg sem mod} and~\eqref{equacao rad compl diverg mod} captures infinite sign changes, almost surely.

\subsection{The case \texorpdfstring{$\sum_{n\leq x}\frac{f(n)}{n^\alpha}$}{sum n <= x (f(n))/(n alpha)}} This case can be treated similarly as the previous one, although we can proceed in a more elementary way as in~\cite{aymone_sign_changes}, that is, Theorem~\ref{teorema Harper} is not needed here.

The proof of Theorems~\ref{teorema principal 1} and~\ref{teorema principal 2} are now complete.

\section*{Acknowledgements} 

This project started while I was a visiting Professor at Aix-Marseille Université, and I am warmly thankful for their kind hospitality. 

\section*{Declaration of interests}
The authors do not work for, advise, own shares in, or receive funds from
any organization that could benefit from this article, and have declared no
affiliations other than their research organizations.

\printbibliography

\end{document}







 





\end{document}

